Author’s response to reviews

Title: Baseline Low ALT Activity, a Marker of Sarcopenia and Frailty, is Associated with Increased Long-Term Mortality after COPD Exacerbations. Historic Prospective Cohort Study.

Authors:

Nir Lasman (nir.lasman@sheba.health.gov.il)
Matan Shalom (matan.shalom@sheba.health.gov.il)
Natia Turpashvili (natia.turpashvili@sheba.health.gov.il)
Gal Goldhaber (gal.goldhaber@sheba.health.gov.il)
Yulia Lifshitz (yulia.lifshitz@sheba.health.gov.il)
Eyal Leibowitz (heartman@matav.net.il)
Gideon Berger (g_berger@rambam.health.gov.il)
Gilat Zalzman-Shenhav (gilat.shenhav-zaltzman@sheba.health.gov.il)
Adi Brom (adi.brom@sheba.health.gov.il)
Dor Cohen (dor.cohen@sheba.health.gov.il)
Chen Avaki (chen.avaki@sheba.health.gov.il)
Gad Segal (gad.segal@sheba.health.gov.il)

Version: 2 Date: 22 Apr 2020

Author’s response to reviews:

Attached please find a point-by-point response to the 2nd review